Check for updates

GOPEN ACCESS

Citation: Klein A, Eggerbauer E, Potratz M, Zaeck LM, Calvelage S, Finke S, et al. (2022) Comparative pathogenesis of different phylogroup I bat lyssaviruses in a standardized mouse model. PLoS Negl Trop Dis 16(1): e0009845. https://doi.org/ 10.1371/journal.pntd.0009845

Editor: Ashley C. Banyard, Animal Health and Veterinary Laboratories Agency: Animal and Plant Health Agency, UNITED KINGDOM

Received: September 22, 2021

Accepted: December 14, 2021

Published: January 18, 2022

Copyright: © 2022 Klein et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: This research was funded by an intramural collaborative research grant on Lyssaviruses at the Friedrich-Loeffler-Institut (Rie-0375) to CMF. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

RESEARCH ARTICLE

Comparative pathogenesis of different phylogroup I bat lyssaviruses in a standardized mouse model

Antonia Klein¹, Elisa Eggerbauer², Madlin Potratz², Luca M. Zaeck², Sten Calvelage³, Stefan Finke², Thomas Müller², Conrad M. Freuling¹*

1 Friedrich-Loeffler-Institut (FLI), Federal Research Institute for Animal Health, Greifswald-Insel Riems, Germany, 2 Institute of Molecular Virology and Cell Biology, Friedrich-Loeffler-Institut (FLI), Federal Research Institute for Animal Health, Greifswald-Insel Riems, Germany, 3 Institute of Diagnostic Virology, Federal Research Institute for Animal Health, Friedrich-Loeffler-Institut (FLI), Greifswald-Insel Riems, Germany

* conrad.freuling@fli.de

Abstract

A plethora of bat-associated lyssaviruses potentially capable of causing the fatal disease rabies are known today. Transmitted via infectious saliva, occasionally-reported spillover infections from bats to other mammals demonstrate the permeability of the species-barrier and highlight the zoonotic potential of bat-related lyssaviruses. However, it is still unknown whether and, if so, to what extent, viruses from different lyssavirus species vary in their pathogenic potential. In order to characterize and systematically compare a broader group of lyssavirus isolates for their viral replication kinetics, pathogenicity, and virus release through saliva-associated virus shedding, we used a mouse infection model comprising a low (10² TCID₅₀) and a high (10⁵ TCID₅₀) inoculation dose as well as three different inoculation routes (intramuscular, intranasal, intracranial). Clinical signs, incubation periods, and survival were investigated. Based on the latter two parameters, a novel pathogenicity matrix was introduced to classify lyssavirus isolates. Using a total of 13 isolates from ten different virus species, this pathogenicity index varied within and between virus species. Interestingly, Irkut virus (IRKV) and Bokeloh bat lyssavirus (BBLV) obtained higher pathogenicity scores (1.14 for IRKV and 1.06 for BBLV) compared to rabies virus (RABV) isolates ranging between 0.19 and 0.85. Also, clinical signs differed significantly between RABV and other bat lyssaviruses. Altogether, our findings suggest a high diversity among lyssavirus isolates concerning survival, incubation period, and clinical signs. Virus shedding significantly differed between RABVs and other lyssaviruses. Our results demonstrated that active shedding of infectious virus was exclusively associated with two RABV isolates (92% for RABV-DogA and 67% for RABV-Insectbat), thus providing a potential explanation as to why sustained spillovers are solely attributed to RABVs. Interestingly, 3D imaging of a selected panel of brain samples from bat-associated lyssaviruses demonstrated a significantly increased percentage of infected astrocytes in mice inoculated with IRKV (10.03%; SD ±7.39) compared to RABV-Vampbat (2.23%; SD±2.4), and BBLV (0.78%; SD±1.51), while only individual infected cells were identified in mice infected with Duvenhage virus (DUVV).

Competing interests: The authors have declared that no competing interests exist.

These results corroborate previous studies on RABV that suggest a role of astrocyte infection in the pathogenicity of lyssaviruses.

Author summary

Globally, there are at present 17 different officially recognized lyssavirus species posing a potential threat for human and animal health. Bats have been identified as carriers for the vast majority of those zoonotic viruses, which cause the fatal disease rabies and are transmitted through infectious saliva. The occurrence of sporadic spillover events where lyssaviruses are spread from bats to other mammalian species highlights the importance of studying pathogenicity and virus shedding in regard to a potentially sustained onward cross-species transmission. Therefore, as part of this study, we compared 13 different isolates from ten lyssavirus species in a standardized mouse infection model, focusing on clinical signs, incubation periods, and survival. Based on the latter two, a novel pathogenicity index to classify different lyssavirus species was established. This pathogenicity index varied within and between different lyssavirus species and revealed a higher ranking of other bat-related lyssaviruses in comparison to the tested Rabies virus (RABV) isolates. Altogether, our results demonstrate a high diversity among the investigated isolates concerning pathogenicity and clinical picture. Furthermore, we comparatively analyzed virus shedding via saliva and while there was no indication towards a reduced pathogenicity of bat-associated lyssaviruses as opposed to RABV, shedding was increased in RABV isolates. Additionally, we investigated neuronal cell tropism and revealed that bat lyssaviruses are not only capable of infecting neurons but also astrocytes.

1. Introduction

The Lyssavirus genus of the family Rhabdoviridae within the order Mononegavirales comprises highly neurotropic, single negative-strand RNA viruses [1], which are capable of causing rabies, an acute and invariably fatal viral encephalitis [2]. At present, 17 lyssavirus species are recognized as separate taxonomic entities [1]. Based on antigenic divergence and phylogenetic relationships, lyssavirus species can be grouped into phylogroups [3]. Phylogroup I include the prototypical rabies virus (RABV), Aravan virus (ARAV), Australian bat lyssavirus (ABLV), Bokeloh bat lyssavirus (BBLV), Duvenhage virus (DUVV), European bat lyssavirus 1 (EBLV-1), European bat lyssavirus 2 (EBLV-2), Gannoruwa bat lyssavirus (GBLV), Irkut lyssavirus (IRKV), Khujand lyssavirus (KHUV), and Taiwan Bat Lyssavirus (TWBLV), while Lagos bat lyssavirus (LBV), Mokola lyssavirus (MOKV), and Shimoni bat lyssavirus (SHIBV) are members of phylogroup II. Based on phylogenetic distance, the most genetically divergent lyssaviruses, including Ikoma virus (IKOV), Lleida bat lyssavirus (LLEBV), and West Caucasian bat lyssavirus (WCBV), have been tentatively classified within a dispersed phylogroup III [4]. Two new lyssaviruses found in Europe, the Kotalahti bat lyssavirus (KBLV) [5], and Africa, the Matlo bat lyssavirus (MBLV) [6], are not yet approved as new virus species. While almost all lyssavirus species are strongly associated with chiropteran hosts [4], RABV is the only lyssavirus maintained in many different species of mesocarnivores around the world. Exceptions include the circulation of RABV in multiple species of bats in the New World and the reported role of a small primate, the marmoset, as an RABV reservoir in Brazil [2,7]. Most bat

lyssaviruses seem to be restricted to a limited number of reservoir host species they have been steadily co-evolving with over time [8].

For RABV, transmission, particularly from vampire bats, to other non-bat mammals is common in the Americas [9]. Nonetheless, sporadic spillover infections of other bat lyssaviruses to non-bat mammal species, including humans, emphasize the threat these viruses pose for both human and animal health [10,11]. In contrast to phylogroup I lyssaviruses, failure of protection against the more divergent phylogroup II and III lyssaviruses has been demonstrated for all commercially available vaccines [3,12–14]. While antigenicity and phylogeny of lyssaviruses have been well studied, the comparative pathogenicity of bat lyssavirus isolates in different species is of scientific interest, particularly against the background of a highly variable pathogenicity between phylogroups observed in mice [15]. Partly, pathogenicity of bat lyssaviruses was tested in ferrets [16], foxes [17–19], raccoon dogs [18,20–22], cats [18,21–28], dogs [18,21,22,24,25,29-32], and skunks [18,21,22,33]. In most of these studies and also when mice were used, either only a limited number of bat lyssavirus species were compared or different viral variants of one particular lyssavirus species have been studied [34,34-40]. Varying experimental designs and conditions often prevent a reliable and broader comparative assessment of the pathogenicity of bat lyssaviruses from these studies. However, in many of the aforementioned studies, the results regarding limited pathogenicity of bat lyssaviruses in nonbat mammals seem to contradict reported bat lyssavirus-borne human fatalities [10].

Moreover, it has not yet been clarified why cross-species transmissions to either humans or animals are more often seen with bat-associated RABVs [8] compared to other bat lyssaviruses for which such events seem to be relatively rare [10]. While the underlying mechanisms of cross-species transmissions are not yet completely understood, virus shedding is assumed to be a key factor, particularly for sustained spillovers [41]. Therefore, it is of importance to understand whether potential spillover hosts are shedding virus and can subsequently transmit it to conspecifics or other mammals, including humans. Such assessment of the potential impact of bat-associated lyssaviruses on public health is particularly challenging but of great importance. To this end, we compared the pathogenicity of 13 phylogroup I lyssaviruses in a standardized mouse model using different inoculation doses and routes. Furthermore, shedding of virus in saliva of infected mice was measured in order to assess the likelihood of onward transmission. Based on obtained pathogenicity data, we further developed a pathogenicity index for comparison and classification of lyssavirus-induced pathogenicity. Since it was shown that the degree of central nervous system resident astrocyte infection differed in RABV field strains compared to laboratory-adapted fixed virus strains, potentially affecting their pathogenicity [42], the cell tropism of a selected lyssavirus panel from diseased mice was analyzed. Even though the restricted number of lyssaviruses analyzed for astrocyte infection hinders a full comparison, the results support previous studies of RABV on the association of astrocyte tropism and pathogenicity.

2. Material and methods

2.1. Ethics statement

The experimental work in mice strictly followed the European guidelines on animal welfare and care according to the authority of the Federation of European Laboratory Animal Science Associations (FELASA) [43]. Animal experiments were evaluated, reviewed, and approved by the animal welfare committee (Landesamt für Landwirtschaft, Lebensmittelsicherheit und Fischerei Mecklenburg-Vorpommern, LALLF M-V/TSD 7221.3–2.1-002/11; M-V/TSD/ 7221.3-2-001/18) and supervised by the commissioner for animal welfare at the Friedrich-

Lab-ID	Name	Virus species	Year	Host	Origin	Accession Number	
5989	RABV-DogA	RABV	2002	dog (<i>Canis lupus familiaris</i>) Azerbaijan		LN879480	
13205	RABV-Raccoon	RABV	1981	raccoon (Procyon lotor) United States		MN862283	
34886	RABV-Vampbat	RABV	1973	vampire bat (<i>Desmodus rotundus</i>) Latin America		PRJEB46947	
13240	RABV-Insectbat	RABV	1986	big brown bat (<i>Eptesicus fuscus</i>) Canada		PRJEB46947	
13027	EBLV-1	EBLV-1	1982	human (Yuli virus) Russia		LT839613	
16618	EBLV-2	EBLV-2	2007	daubenton's bat (<i>Myotis daubentonii</i>) Germany		KY688138	
29008#	BBLV	BBLV	2010	natterer's bat (Myotis Nattererii) Germany		KF245925	
46579	ARAV	ARAV	1991	lesser mouse-eared bat (<i>Myotis blythii</i>) Kyrgyzstan		EF614259	
46580	KHUV	KHUV	2001	whiskered bat (Myotis mystacinus)	Tajikistan	NC025385	
46582	IRKV	IRKV	2002	greater tube-nosed bat (<i>Murina leucogaster</i>) Russia		NC020809	
39663	GBLV	GBLV	2016	indian flying fox (Pteropus medius) Sri Lanka		KU244267	
12862	DUVV	DUVV	1971	human South Africa		EU293119	
13849	ABLV	ABLV	1986	human Australia		AF418014	

Table 1. Viruses used in the study, including details of their year of isolation, the respective host, and origin.

[#]A recombinant virus was used.

https://doi.org/10.1371/journal.pntd.0009845.t001

Loeffler-Institut (FLI) representing the Institutional Animal Care and Use Committee (IACUC).

2.2. Viruses and cells

A total of 13 virus isolates representing ten different phylogroup I lyssaviruses originating from Europe (EBLV-1, EBLV-2, BBLV), Asia (ARAV, KHUV, IRKV, GBLV), Africa (DUVV), Australia (ABLV), and the Americas (RABV) were included in this study. Regarding the latter, two bat-related RABV isolates, one from an insectivorous (RABV-Insectbat) and one from a hematophagous bat (RABV-Vampbat) (Table 1), were selected. For EBLV-1, DUVV, and ABLV, isolates from human cases were used. The Asian bat lyssaviruses ARAV, IRKV, GBLV and KHUV were kindly provided by the Animal Plant and Health Agency (APHA), Weybridge, United Kingdom through the European Virus Archive global (EVAg). All other isolates originated from the virus archive of the FLI, Riems, Germany. For comparison, two representatives of classical non-bat RABVs, one being an isolate from a dog from Azerbaijan (RABV--DogA) and the other being a wildlife variant isolated from a raccoon from North America (RABV-Raccoon), both of which had been used in previous infection studies in raccoons [20], were included. Cell lines used in this study were obtained from the Collection of Cell Lines in Veterinary Medicine (CCLV; FLI, Riems, Germany). Mouse neuroblastoma cells (Na 42/13, CCLV-RIE 0229) maintained in a mixture of equal volumes of Eagle MEM (Hanks' balanced salts solution) and MEM (Earle's balanced salts solution) medium supplemented with 10% fetal bovine serum and penicillin/streptomycin (100 U/ml and 100 µg/ml, respectively) were used for propagation of virus stocks, titration, viral replication kinetics, and virus isolation.

2.3. Full genome sequencing

All viruses taken from archived samples were subjected to full genome sequencing essentially as described before [44]. Briefly, RNA was automatically extracted on a KingFisher Flex platform (Thermo Fisher Scientific, Waltham, MA, USA) using the RNAdvance Tissue Kit (Beckmann Coulter, Brea, CA, USA). Double stranded cDNA was generated from 350 ng total RNA using the SuperScript IV First-Strand cDNA Synthesis System (Invitrogen/Thermo Fisher Scientific, Waltham, MA, USA) and the NEBNext Ultra II Non-Directional RNA Second Strand Synthesis Module (New England Biolabs, Ipswich, MA, USA). After conversion into cDNA, fragmentation was achieved by ultrasonication on a Covaris M220 (Covaris, Brighton, UK). Subsequently, Ion Torrent-specific sequencing libraries were generated using the GeneRead L Core Kit (Qiagen, Hilden, Germany) together with IonXpress barcode adaptors (Thermo Fisher Scientific). After quantification (QIAseq Library Quant Assay Kit, Qiagen) and quality control (2100 Bioanalyzer, High sensitivity DNA Kit, Agilent Technologies, Santa Clara, CA, USA) of the libraries, sequencing was performed on an Ion Torrent S5XL instrument utilizing Ion 530 chips and reagents according to the manufacturer's instructions.

2.4. Viral propagation and replication kinetics

In order to generate virus stocks for inoculation, Na 42/13 cells were infected at a multiplicity of infection (MOI) of 0.001, and incubated at 37° C and 5% CO_2 . Depending on the viral strain, supernatants were harvested 72 to 168 hours post-infection (hpi) when 100% of the cell monolayer was infected. Infection was assessed using a control dish stained with a fluorescein isothiocyanate (FITC)-conjugated monoclonal antibody mix (SIFIN, Germany/Fujirebio, Belgium). For growth curves, Na 42/13 cells were infected at an MOI of 0.001. Cell culture supernatants were harvested at 0, 16, 24, 48, 72, 96, and 168 hpi. The virus titers (tissue culture infective dose 50—TCID₅₀) were determined by endpoint titration of three technical replicates and subsequent calculation by the Spearman-Karber method [45].

2.5. Animal experiments

For the experimental studies, three- to four-week-old BALB/c mice from a commercial breeder (Charles River, Germany) were used. Animals were randomly assigned to groups and housed in individual, labelled cages with water and food provided ad libitum. Per lyssavirus isolate, six mice each were inoculated intramuscularly (i.m.) in the femoral muscle using a high (10^5) $TCID_{50}/30 \mu$) and a low (10² $TCID_{50}/30 \mu$) viral dose. Additionally, one group of six and one of three mice was inoculated intranasally (i.n.) with $10^2 \text{ TCID}_{50}/10 \,\mu\text{l}$ and intracranially (i.c.) with $10^2 \text{ TCID}_{50}/30 \mu$ l, respectively. Mock-infected control groups for each administration route were inoculated with 10 μ l or 30 μ l of cell culture medium respectively. Animals were monitored for 21 days post-infection (dpi). Body weight and clinical signs were recorded daily for each animal using clinical scores (S2 Table). If animals showed more than one clinical sign at a given time point, the most prominent one dominating the physical condition was recorded. Once mice developed clinical signs, they were checked twice a day. Animals were humanely euthanized at the humane endpoint or after 21 days by cervical dislocation under anesthesia with isoflurane (Isofluran CP, CP Pharma, Germany). Immediately before euthanasia, oropharyngeal swabs were taken of all mice that succumbed to infection using dry sterile cotton swabs (Nerbe plus GmbH & Co. KG, Germany), which were placed into 1500 µl of cell culture medium supplemented with penicillin/streptomycin as described above. Upon euthanasia, salivary glands and brain samples of all animals were taken and stored at -80° C until further processing.

2.6. Diagnostic assays

Presence of lyssavirus antigen in heat-fixed brain tissue samples was detected by direct fluorescence antibody test (DFA) using FITC-conjugated monoclonal antibodies (SIFIN, Germany and Fujirebio, Belgium) as well as defined positive and negative controls [46].

Brain, salivary gland and oropharyngeal swab samples were used to detect lyssaviral RNA. Briefly, organ samples were homogenized in 1000 μ l cell culture medium using a TissueLyser (Qiagen, Germany) with a 3 mm steal bead. Homogenates as well as oral swabs were

centrifuged at 3750 x g for 10 minutes. Viral RNA was then extracted from the supernatant (100 µl) using the NucleoMagVet kit (Macherey&Nagel, Germany) according to the manufacturer's instructions in a KingFisher/BioSprint 96 magnetic particle processor (Qiagen, Germany). Viral RNA was detected by an RT-qPCR targeting the N-gene using the R14-assay (RABV-, EBLV-1-, EBLV-2-, and BBLV) and the R13-assay (ABLV, DUVV) [47,48]. For GBLV, primers and probe were specifically designed (S1 Table) and the protocol was run separately but with the same conditions as below. The PCR mastermix was prepared using the AgPath-ID One-step RT-PCR kit (Thermo Fisher Scientific, USA) in a volume of 10 μl including 0.5 μ l of β -Actin-mix2-HEX as internal control and 2.5 μ l of extracted RNA. The reaction was performed for 10 minutes at 45° C for reverse transcription and 10 minutes at 95° C for activation, followed by 42 cycles of 15 seconds at 95° C for denaturation, 20 seconds at 56° C for annealing and 30 seconds at 72° C for elongation. Fluorescence was measured during the annealing phase. RNA specific for ARAV, IRKV and KHUV was detected using the pan-lyssa realtime PCR targeting both the N- and L-gene [47,48]. Here, the RT-qPCR reaction was prepared using the OneStep RT-PCR kit (Qiagen, Germany), adjusted to a volume of 12.5 µl with an internal control mastermix, based on β -Actin, running in parallel. The reaction consisted of 10 minutes at 45° C for reverse transcription and 10 minutes at 95° C for activation, followed by 45 cycles of 15 seconds at 95° C for denaturation, 20 seconds at 56° C for annealing and 30 seconds at 72° C for elongation. All RT-qPCRs were performed on a BioRad CFX96 Real-Time System (Bio-Rad, USA).

RT-qPCR positive salivary glands and oral swab samples were subjected to virus isolation in cell culture using the rabies tissue-culture infection test (RTCIT) [49]. Briefly, either the respective swap sample or supernatant from the homogenized organ suspension prepared using cell culture media as described above and supplemented with 10% fetal bovine serum and 200 U/ml und 200 µg/ml penicillin/streptomycin, was mixed with dextran-pretreated Na 42/13 cell suspension at an equal ratio. The mixture was then incubated at 37° C and 5% CO₂ for 30 minutes and centrifuged at 1250 x g for 10 minutes. The obtained cell pellets were resuspended in T25 cell culture flasks and incubated for three to four days at 37° C and 5% CO₂. A control-dish was fixed, stained with a commercial FITC-conjugated monoclonal antibody conjugate (SIFIN, Germany/Fujirebio, Belgium), washed and microscopically analyzed for the presence of virus. Three consecutive serial passages were used to confirm a negative result.

2.7. Antibodies for immunofluorescence imaging of solvent-cleared brain sections

To detect bat lyssavirus antigen in infected brains, a polyclonal rabbit serum against recombinant RABV P protein (P160-5, 1:3,000 in PTwH [0.2% Tween 20 in PBS with 10 µg/ml heparin]) was used [50]. The following commercial primary antibodies were used: chicken anti-GFAP (Thermo Fisher, USA; #PA1-1004, RRID:AB_1074620, 1:1,500 in PTwH) and guinea pig anti-NeuN (Synaptic Systems, Germany; #266004, RRID:AB_2619988, 1:800 in PTwH). Donkey anti-rabbit Alexa Fluor 568 (Thermo Fisher, USA; #A10042, RRID:AB_2534017) and donkey anti-guinea pig Alexa Fluor 647 (Dianova, Germany; #706-605-148, RRID: AB_2340476) were used as secondary antibodies, each at a dilution of 1:500 in PTwH.

2.8. Immunostaining and clearing of brain tissue samples

Immunostaining and clearing protocols from previous reports [51–53] were modified and performed as described previously [54]. All incubation steps were performed on an orbital shaker. Briefly, the brains fixed with paraformaldehyde (PFA) were cut into 1 mm thick slices using a vibratome (Leica Biosystems, Germany, VT1200S). After bleaching with 5% H_2O_2/PBS overnight at 4° C the samples were pre-permeabilized (0.2% Triton X-100/PBS) twice for 3 h each at 37° C and then permeabilized (0.2% Triton X-100/20% DMSO/0.3 M glycine/PBS) for 2 days at 37° C. After subsequent blocking (0.2% Triton X-100/10% DMSO /6% donkey serum/PBS) at 37° C for further 48 h, primary antibodies diluted in 3% donkey serum/5% DMSO/PTwH were added for a total of 7 days at 37° C. After 3.5 days, the antibody solution was renewed once. Subsequently, the samples were washed with PTwH four times with increasing intervals, leaving the final wash on overnight. Secondary antibodies were diluted in 3% donkey serum/PTwH and incubation was performed as described for the primary antibodies. After further washing with PTwH four times, leaving the final wash on overnight, the samples were dehydrated in a graded ethanol series (30%, 50%, 70% in aqua ad iniectabilia [pH 9–9,5] and twice 100%; each for ≥ 6 h) at 4° C. Subsequently, they were delipidated for 2 h in *n*-hexane at room temperature. Gradually replacing the *n*-hexane with ethyl cinnamate (ECi), they were then incubated in ECi until optically transparent. For confocal laser-scanning microscopy analysis, the cleared samples were embedded in 3D-printed imaging chambers as described before [55].

2.9. Confocal laser-scanning microscopy and image processing

Confocal z-stacks were acquired with a Leica DMI 6000 TCS SP5 confocal laser-scanning microscope equipped with a $40 \times /1.10$ water immersion HC PL APO objective using the Leica Application Suite Advanced Fluorescence software (v2.7.3.9723). Fluorescence was acquired sequentially between lines with a pinhole diameter of 1 Airy unit and a z-step size of 0.5 μ m.

The quantification of infected neurons and astrocytes in 1 mm thick brain sections was done as according to previous description [42]. To this end, confocal image stacks were split into individual channels using Fiji, an ImageJ (v1.52h) distribution package [56]. After bleach correction (simple ratio, background intensity 5.0) brightness and contrast were adjusted for each channel. The 3D objects counter plugin was used to identify objects in each channel [57]. The resulting objects map was then overlaid with the RABV P channel to detect and count infected objects. Visualization was done using arivis vision4D (v3.4.0).

2.10. Statistical analysis

Survival of mice was displayed in Kaplan-Meier curves and statistically analyzed by log-rank (Mantel-Cox) test. Significant differences in the means of incubation periods and astrocyte tropism were assessed using ordinary one-way ANOVA with Tukey's multiple comparison test. All statistical analyses were performed using Prism version 8 (GraphPad Software, La Jolla, USA).

2.11. Calculation of intramuscular pathogenicity index (IMPI)

For the calculation of the IMPI, we followed the example of the intracerebral pathogenicity index (ICPI) test for Newcastle disease virus [58] with the following modifications: The clinical observations of individual animals recorded every 12 to 24 hours over a period of up to 21 days were transferred to a daily rating scheme. Mice were scored as follows: 0 if healthy; 1 if sick, and 2 if dead. Dead individuals were scored as 2 at each of the remaining daily observations after death. For calculation of the IMPI, only animals from the i.m.-inoculated groups were included. The intramuscular pathogenicity index was then calculated based on the following formula: cumulative score for sick animals + cumulative score of dead animals / 126 (number of animals x days of observation, i.e. 6 x 21). The index is determined as the mean score per mouse over a 21-day-period, i.e. very pathogenic viruses showing high and less pathogenic ones showing lower indices. A minimum index value of 0 corresponds to absolutely

apathogenic animals throughout the complete observation period while a maximum index value of 2 would be reached if all mice died at day 1 post-infection.

3. Results

3.1. In vitro replication kinetics

In mouse neuroblastoma cells (Na 42/13), the tested lyssaviruses replicated to maximum titers ranging from 10^5 TCID₅₀/ml (ABLV) to $10^{7.75}$ TCID₅₀/ml (RABV-Vampbat) after 168 hours (Fig 1). RABV-Vampbat, IRKV, GBLV, RABV-DogA, and EBLV-1 exhibited titers around 10^7 TCID₅₀/ml and higher, while the titers for the rest of the isolates ranged between 10^5 and 10^6 TCID₅₀/ml. IRKV and EBLV-1 showed the fastest replication with measurable titers obtained already after 16 and 24 hpi, while all other isolates yielded measurable titers either after 48 or 72 hpi. The RABV-Raccoon variant in particular exhibited a slow replication kinetic; replication started after 72 hpi but with comparably low titers even after 96 hpi (Fig 1).

3.2. Survival rates

The survival rates of mice investigated according to the aforementioned experimental setup $(\underline{Fig 2})$ considerably differed depending on the lyssavirus, the route of infection, and the



Fig 1. In vitro replication kinetics of lyssavirus isolates in Na 42/13 cells infected with a MOI of 0.001. The mean and standard errors of three replicates are indicated.

https://doi.org/10.1371/journal.pntd.0009845.g001



Fig 2. Experimental setup. Outline of the animal experiment and sample collection.

infectious dose (Fig 3). In the low dose i.m. groups, survival rates varied between 100% (RABV-Raccoon, EBLV-1) and 0% (BBLV) (Fig 3A). All mice i.m. infected with a high dose of ARAV, GBLV, IRKV, RABV-DogA and BBLV developed clinical signs and had to be euthanized. In contrast, 67% and 50% of mice survived following high dose inoculation with ABLV and RABV-Raccoon, respectively, and all animals survived after DUVV infection (Fig 3B). In contrast, when comparing bat lyssaviruses with classical RABV strains, there was no significant difference in survival rates among i.m. low dose (p = 0.0862, Log-rank (Mantel-Cox) test) and high dose (p = 0.8761, Log-rank (Mantel-Cox) test) infected groups. All mock-infected control mice did not develop any clinical signs and survived until the end of the observation period. All mice inoculated i.c. with the different isolates developed clinical signs and were euthanized, except one mouse inoculated with BBLV, which survived until the lyssavirus isolates varied between 33%, (EBLV-1, EBLV-2 and BBLV) and 100%, (RABV-Raccoon, RABV-Insectbat, RABV-Vampbat and IRKV, S1B Fig).

3.3. Incubation periods

In the i.m. low dose groups, the longest incubation periods were observed for mice inoculated with the reference strain RABV-DogA (mean: 17 days, SD±3) and with RABV-Vampbat (mean: 17 days, SD±0). However, in the group inoculated with RABV-Vampbat only a single animal developed clinical signs at all, the same applied to mice inoculated with DUVV.





Incubation periods of mice inoculated with BBLV (mean: 10 days, SD±1), ARAV (mean: 9 days, SD±1.8), IRKV (mean: 8 days, SD±0.5) were significantly shorter compared to RABV--DogA (Fig 4A). In contrast, all mice infected with RABV-Raccoon and EBLV-1 survived until 21 dpi showing no clinical signs at all. Differences in incubation periods in the i.m. high dose groups were more pronounced. Mice inoculated with RABV-DogA (mean: 7 days, SD±0.8),



Fig 4. Incubation periods after low dose (A) and high dose (B) i.m. infection. Mean values are provided as horizontal lines. Statistical differences between the mean of RABV-DogA as a reference challenge strain and the means of other lyssaviruses are indicated (* $p \le .05$; ** $p \le .01$; *** $p \le .001$; ordinary one-way ANOVA with Tukey's multiple comparison test).

EBLV-1 (mean: 7 days, SD±1.5), BBLV (mean: 9 days, SD±0.8), IRKV (mean: 6 days, SD±0.4) and GBLV (mean: 8 days, SD±2.3) displayed clinical signs earlier (mean <10 days) than the remaining groups (Fig 4B). Regarding classical and bat-associated RABVs, the mean incubation for RABV-DogA was significantly different to those for RABV-Raccoon (mean: 13 days, SD±5.8) and RABV-Vampbat (mean: 12 days, SD±2.8) (Fig 4B). None of the mice infected with DUVV (high dose; i.m.) showed clinical signs until the end of the observation period.

3.4. Clinical signs

Independent of either the bat lyssavirus isolates or the classical RABV variants used, mice that succumbed to infection displayed clinical pictures suggestive of rabies. In general, clinical signs summarized as a progressive deterioration of the general health condition included decreased activity, hunched back, ruffled fur, weight loss, and lethargy or loss of alertness. Other clinically evident signs comprised paralysis, paresis, spasms, convulsive seizures, tremor, pruritus, aggressiveness, tameness, moving in circles, or extremely increased uncoordinated movements (Fig 5). For i.m. infected mice, disease progression after onset of clinical signs was either peracute with rapid development of clinical signs within <12 hours from healthy to apathetic, or disease duration was comparatively slower starting with mild, unspecific signs evolving into the full clinical picture within two to three days. The former was more frequently observed in mice inoculated with IRKV or KHUV and the latter was particularly common in mice inoculated with BBLV.

After i.m. infection, 41% of diseased mice had a deterioration of their general condition as the most prominent clinical sign, while 52% showed paralysis and paresis. The latter was more pronounced in GBLV (100%), ARAV (91%), EBLV-2 (89%) and IRKV (80%), whereas EBLV-1 and DUVV are outliers in the non-RABV lyssaviruses with no paresis/paralysis observed in diseased mice. Generally, with 63% over 25%, paresis/paralysis was significantly increased in non-RABV after i.m. infection of mice ($\chi^2 = 21.16$, p < 0.0001). Also, paresis/paralysis occurred generally less often in i.c. (11%)- and i.n. (14%)-infected mice (Fig 5).

No clinical signs observed	
Deterioration of general condition	
Paralysis, Paresis	
Spasms, Seizures	
Aggressiveness/Tameness	
Tremor	
Pruritus	
Running in circles	
Hyperactivity	

Inoculation Route	RABV DogA	RABV Raccoon	RABV Insectbat	RABV Vampbat	EBLV-1	EBLV-2	BBLV	GBLV	ABLV	DUVV	IRKV	ARAV	κηυν
1.m.													
i.c.													
i.n.													



3.5. Index-based comparative pathogenicity

To allow for a ranking in pathogenicity, we implemented a novel intramuscular pathogenicity index (IMPI), which takes clinical signs and deaths/euthanasia of all i.m.-inoculated (dose-independently) mice into account. If all infected mice died at day 1, the IMPI would be 2, whereas it would be 0 if all mice survived with no clinical score. Depending on the lyssavirus used, indices ranging between 1.14 and 0.07 were obtained (Fig 6). The IMPI scored highest for IRKV and BBLV (>1) compared to the classical RABV-DogA (0.85). In contrast, RABV--Raccoon and DUVV had the lowest score (<0.19) (Fig 6).

3.6. Infection of neuron and astrocytes by selected bat lyssaviruses

Since pathogenic RABV have recently been shown to exhibit a specific astroglia tropism [42], we analyzed whether other lyssaviruses could infect central nervous system resident astrocytes to a comparable extent and whether differences in the pathogenicity correlate with astroglia infection levels. Therefore, the brain cell tropism of selected bat lyssaviruses with a high (IRKV, BBLV) and a low (RABV-Vampbat, DUVV) pathogenicity index (Fig 6) was investigated by 3D immunofluorescence imaging and quantitative analysis of infected neurons and astrocytes (Fig 7).





Intramuscular infections with IRKV, RABV-Vampbat, BBLV and DUVV resulted in a mean of 15.77% (Standard Deviation, SD±8.54), 19.83% (SD±12.59), 7.45% (SD±6.05), and 16.4% (SD±8.81) virus-positive neurons, respectively, demonstrating that IRKV and RABV--Vampbat had significantly higher levels of neuron infection compared to BBLV in clinically diseased mice (Fig 7B and 7C). Concerning astroglia infection, IRKV-infected mice featured the highest percentage of infected astrocytes (10.03%; SD±7.39). While astrocyte infection was lower in RABV-Vampbat (2.23%; SD±2.4)- and BBLV (0.78%; SD±1.51)-infected mice, only individual infected cells were identified in the DUVV-infected mice (Fig 7B and 7C).

3.7. Virus shedding-detection of viral RNA and viable virus in salivary glands and oral swabs

The detection of viral RNA in salivary glands and oral swabs in mice differed depending on the lyssavirus species used for inoculation. Positivity rates for all salivary gland and oral swab samples were highest for RABV-DogA (100%), followed by RABV-Insectbat (92%),



С

		Neuro	ns		Astrocytes					
Virus	Counted	Infected	Mean (%)	SD	Counted Infected Mean (%) SD					
IRKV	14139.00	2121.00	15.77	8.54	8624.00 748.00 10.03 7.39					
RABV Vampbat	12068.00	2338.00	19.83	12.59	9429.00 178.00 2.23 2.40					
BBLV	17487.00	1295.00	7.45	6.05	11210.00 61.00 0.78 1.51					
DUVV	5115.00	759.00	16.40	8.81	3560.00 3.00 0.08 0.18					

Fig 7. Comparison of the astrocyte tropism of different bat lyssaviruses with a high and low IMPI. A) Indirect immunofluorescence of solvent-cleared brain sections demonstrates the presence of lyssavirus phosphoprotein P (red), neurons (blue, marker: NeuN) and astrocytes (green, marker: glial fibrillary acidic protein, GFAP). Insets show RABV P accumulation (red) at GFAP-positive cells (green). x, y = 387.5 µm, 387.5 µm; z = 65.5 µm (IRKV), 36.5 µm (RABV-Vampbat), 64.5 µm (BBLV), 65.5 µm (DUVV). Scale bar = 100 µm (overview), 15 µm (inset). **B)** Percentage of infected astrocytes (black dots) and neurons (gray squares). Per virus, 3 to 11×10^3 astrocytes and 5 to 17×10^3 neurons were counted in independent confocal z-stacks in two animals per isolate (one animal for DUVV). Each dot represents the ratio of infected astrocytes or neurons in an analyzed z-stack. Mean values are provided as horizontal lines. Statistical comparisons of the means between the different groups are indicated for those with a statistically significant difference (* $p \le .05$; ** $p \le .01$; *** $p \le .00$; ordinary one-way ANOVA with Tukey's multiple comparison test). **C)** Corresponding data table for the measurements.

https://doi.org/10.1371/journal.pntd.0009845.g007

RABV-Raccoon (33%) and RABV-Vampbat (33%) (Fig 8A). Independent of the route of infection, 97% of diseased mice inoculated with RABV strains were positive for viral RNA in salivary glands, and in 72% of mice viral RNA was also detected in the corresponding oral swabs (Fig 8B). In contrast, mice infected with non-RABV bat lyssaviruses exhibited significantly lower positivity rates (p<0.0001, Fischer's exact test), i.e. in only 50% of the diseased animals, salivary glands were positive for viral RNA, and 12% exhibited both virus RNA-positive salivary glands and oral swabs (Fig 8B). Regarding the presence of infectious virus, a similar pattern was observed when all routes of infection were considered (Fig 8C and 8D). When grouped together, infectious virus could be isolated in 86% of salivary gland samples from mice that succumbed to RABV, while infectious virus shedding, as determined by virus isolation from both salivary glands and oral swabs, was 47% overall, with the highest proportion in



Fig 8. Comparison of virus shedding in lyssavirus-infected mice. Percentage of animals positive/negative for viral RNA (A, B) and viable virus (C, D) in either salivary glands or oral swabs or both according to individual lyssaviruses (A, C) or grouped according to RABVs and non-RABV bat lyssaviruses (B, D). **Correlation between ct-values as obtained in RT-qPCR and results of virus isolation in salivary glands (E) and oral swabs (F).** Here, only animals were considered were active shedding (positive salivary gland and positive corresponding oral swab) was observed. Individual ct-values are shown and the mean is indicated. Successful virus isolations in cell culture are highlighted.

RABV-DogA (92%) and RABV-Insectbat (67%). While BBLV- and GBLV-diseased mice had positive salivary glands at frequencies of 83% and 67%, respectively (Fig 8C), no virus was isolated from oral swabs. None of the salivary glands or oral swabs from ARAV-, IRKV-, and KHUV-infected mice were positive for viral RNA or infectious virus (Fig 8A and 8C). Combined, similar to the presence of viral RNA, mice infected with non-RABV bat lyssaviruses exhibited significantly lower (p<0.0001, Fischer's exact test) positivity rates for infectious virus isolation as compared to RABV-infected mice (Fig 8D).

When data were analyzed according to inoculation dose and route, shedding of infectious virus was observed more often in mice diseased after i.c. inoculation (51%), followed by i.m. high dose (41%), i.n. (35%), and i.m. low dose (29%). The mean ct-values were significantly lower in samples with successful virus isolation from salivary glands (p = 0.003, unpaired t-test, Fig 8E) and corresponding oral swabs (p = 0.003, unpaired t-test, Fig 8F) as opposed to unsuccessful virus isolation. This observation is mainly driven by the low ct-values observed in RABV-Insectbat and RABV-DogA.

4. Discussion

Although lyssaviruses comprise a genetically close group of viruses, which all cause the disease rabies, differences in their phenotype may indicate different risks for veterinary and public health. In our assessment, 13 different lyssaviruses exhibited differences in their replication kinetics in terms of the growth dynamic and the final virus titer. As such, *in vivo* replication kinetics may reflect replication in the animal host and thus may explain differences seen in incubation periods and pathogenicity among individual lyssaviruses. Interestingly, relatively short incubation periods (IRKV, EBLV-1) (Fig 4A and 4B) or relatively high (IRKV, BBLV) or low pathogenicity indices (ABLV) (Fig 6) in the animal model also correlate with the replication kinetics (Fig 1).

Similarly, the number of animals that survived infection with the individual lyssaviruses varied (Fig 3A and 3B). Survival was not associated with belonging to classical RABV as opposed to non-RABV bat lyssaviruses. There was considerable variation seen in incubation periods across lyssavirus species as well as between animals of individual groups (Fig 4A and 4B). However, we have no evidence that bat-associated lyssaviruses cause longer incubation periods in mice compared to observations made in epidemiological bat models [59,60], and supporting case studies [61,62].

The perception and subsequent recording of clinical signs might be biased due to the clinical score being applied for animal welfare reasons and due to the temporally restricted observation scheme. However, all investigated lyssaviruses caused a clinical picture that led to euthanisia or death in mice, albeit at different scale. We observed the clinical outcome to be predominantly dependent on the virus species but also on the route of application. The fact that clinical signs such as paralysis and spasms were more pronounced after i.m. inoculation corroborates previous findings [34]. On the other hand, the observation that mice infected i. m. with non-RABV bat lyssaviruses were more likely to develop spasms and paralysis is interesting but requires further investigation. While lyssavirus species-dependent clinical signs were reported before [36], in summary, no clear pattern in regard to particular lyssavirus species was evident in our study (Fig 5).

Overall, the variation in pathogenicity factors shown in our study highlights the complexity and difficulty to establish a holistic concept for the classification of lyssaviruses. To integrate these factors from our standardized in vivo model, we applied an intramuscular pathogenicity index for lyssaviruses, and thereby demonstrated a high diversity across phylogroup I lyssaviruses that had not been shown to such an extent before. Remarkably, non-RABV bat lyssavirus isolates are among those with the highest IMPI (Fig.6), questioning previous suggestions that bat-related lyssaviruses are less pathogenic [63]. Interestingly, IRKV demonstrated the highest pathogenicity in a ferret model as well in comparison to the other bat-associated lyssaviruses KHUV and ARAV [13]. Moreover, the fact that current RABV-based biologicals provide only partial protection against IRKV challenge [13,64] further emphasizes the higher risk for a lethal outcome associated to IRKV infections. Even though EBLV-1, DUVV, and IRKV have all caused human rabies cases [10,65], no elevated pathogenicity was found in our model using isolates from human cases (EBLV1, DUVV) (Fig.6). Of note, the Yuli isolate we used here is the only available EBLV-1 isolated from a fatal human case [66].

Using 3D high-resolution confocal laser-scanning microscopy, the CNS cell tropism for bat-related lyssaviruses was analyzed here for the first time and provides new insights into their capability to infect astrocytes (Fig 7). A comparison between RABV field isolates and labadapted strains revealed that astrocyte infection after i.m. inoculation is associated with field strains and, thus, might be a potential pathogenicity determinant [42]. In our analyses using representatives of bat-associated lyssaviruses, mice infected with IRKV, as a representative for high pathogenicity (IMPI = 1.14), had significantly higher proportions of infected astrocytes than RABV-Vampbat (IMPI = 0.41)- and BBLV (IMPI = 1.06)-infected mice. Interestingly, only sporadic astrocyte infection was found in DUVV (IMPI = 0.07)-infected mice. Even though the restricted number of lyssaviruses analyzed for astrocyte infection hinders a full comparison, the results support previous studies of RABV on the association of astrocyte tropism and pathogenicity. Additional virus isolates and strains have to be analyzed in further studies to confirm the role of astrocyte tropism in lyssavirus pathogenicity. However, different virus kinetics and astrocyte-related innate immune reactions may affect the progression kinetics, immune pathogenicity, and further spread of the virus to peripheral salivary glands. The latter may represent a key issue in terms of virus transmission and maintenance in host populations. In our analyses, virus shedding was not demonstrated in IRKV-infected mice. In general, virus shedding represents a striking discrepancy between RABV and other bat lyssaviruses, as virus shedding was significantly reduced in non-RABV bat lyssaviruses in the mouse model. Interestingly, while shedding was highest for the dog rabies strain RABV-DogA, bat-related RABV isolates and a raccoon RABV variant demonstrated a lower percentage of active shedding (Fig 8A–8D). Of note, the raccoon RABV lineage in the Americas is a result from an ancient sustained spillover event from bats [67].

By our definition, active shedding is the successful virus isolation from salivary glands and the respective oral swab sample. Of note, a positive result from an oral swab may not necessarily correspond to viral excretion in saliva itself because the swab could also contain desquamated cells, including infected neurons, which may not necessarily have been excreted naturally. However, technically it was not possible to extract only saliva and the same methodology was used for all isolates. Discrepant results in virus isolation in salivary gland samples and corresponding oral swabs can be explained by the fact that neurons innervating the glands are infected, without excreting virus in the lumen of the gland. The reason for the limited shedding of bat-associated lyssaviruses may be a yet unknown block or barrier in virus distribution in the salivary gland of non-bat mammals. Another explanation could be intermittent shedding, i.e. samples might have been taken at time points when virus was temporarily not shed, but the salivary gland was still virus positive. Nevertheless, intermittent shedding does also apply for RABV strains and consequently it does not completely explain the differences. For some isolates the disease duration in mice was very short so that the centripetal spread of virus may not have reached the salivary gland before death or euthanasia. Interestingly, neither viral RNA nor infectious virus could not be found in salivary glands or oral swabs for the three virus species IRKV, KHUV, and ARAV. Furthermore, there was a gradient for virus shedding

with lowest percentage of shedding in low dose i.m. infected animals, and highest in i.c. inoculated ones. While the mean ct-value in successfully isolated samples was significantly lower, there was a high range in ct-values between 20 and 37 (Fig 8E and 8F), and no threshold for successful isolation could be defined.

In any case, our shedding results could mimic the capability of transmission from a spillover host to another conspecific as the prerequisite of a sustained spillover [41], and thereby contribute to an overall risk assessment for lyssaviruses. Our shedding results in the mouse model confirm previous assumptions based on field observations that the potential for sustained spillovers is highest in RABV as opposed to other bat lyssaviruses [8]. However, a further component of successful virus transmission is the initial shedding potential in the respective reservoir host. Animal models have greatly improved our understanding of virushost interactions. When studying virus-host interactions of bat-related lyssaviruses, experimental studies in bats as their primary reservoir hosts would be ideal. While the results of those studies can infer the pathobiology in the host, they are particularly challenging due to their demanding housing conditions, difficult handling, conservation issues and partly strict protection status [68]. Furthermore, they are less suitable to assess the pathogenicity in nonreservoir hosts. Therefore, for reasons of comparison, the selection of reliable and standardized alternative animal models is of great importance in the study of pathogenicity factors. While mice are considered a standard model for studying lyssavirus pathogenicity [69], in fact little attention has been paid to comparability. Here, our model facilitates a direct comparison of pathogenicity data of 13 different phylogroup I bat lyssaviruses by using a consistent approach regarding experimental conditions, e.g. cells for viral propagation, mouse breed, viral doses, inoculation routes as well as observation times and scoring schemes. We thereby optimized an in vivo mouse model established for EBLV-1 [34] by including data on virus shedding. Furthermore, we established a novel matrix for comparing pathogenicity based on clinical parameters, the intramuscular pathogenicity index IMPI. Such indices are commonly used for avian influenza viruses and Newcastle disease viruses to directly infer the potential for causing disease in animals and thus relate to the respective veterinary control measures. In our case for lyssaviruses, the index can also be used to summarize the pathogenic potential of the individual lyssavirus isolate.

We included i.n. and i.c. inoculation routes in our assessment (S1 Fig), with i.c. primarily used as a positive infection control with the low dose virus inoculum, whereas i.n. application was tested as it was speculated before that virus transmission via aerosols could contribute to disease spread among bats and in spillover infection [68,70]. While i.n. application led to infection of some animals (S1B Fig) likely via the olfactory pathway [71], no clear indication for a specific role in bat lyssaviruses was found (Fig 5), supporting experimental studies in bats [72,73]. Therefore, we then focused our analyses on pathogenicity to i.m. inoculation as the most likely route of virus infection. For virus shedding, we wanted to assess the virus' potential to be transmitted when an animal develops disease, and therefore all inoculation routes were considered. Nevertheless, there are some limitations in our study. The number of mice used was kept to a minimum, respecting animal welfare guidelines on the 3R principle [74]. Also, propagation of viruses in cell culture was a necessary requirement to obtain sufficient virus stocks for the experiments. To minimize the possibility of adaption to cell culture that may influence the results obtained, we used Na 42/13 cells, a mouse neuroblastoma cell line considered to be primary target cells for lyssaviruses, and kept the number of passages as low as possible. Also, the likelihood of adaptive mutation is regarded low since lyssaviruses, i.e. RABV [75] and EBLV-1 [76] have among the lowest mutation rates of RNA viruses [77]. The full genome sequences derived from passaged material did not give evidence of nucleotide exchanges compared to other sequences of the primary isolates.

5. Conclusion

Here, we have comparatively assessed the pathogenicity and virulence of a wide diversity of lyssaviruses belonging to phylogroup I using a standardized and systematic approach. Interestingly, we found in our model bat-associated lyssaviruses, which are more pathogenic and virulent than a classical RABV challenge strain. In fact, no tendency towards a generally reduced pathogenicity of bat-associated lyssaviruses as opposed to classical RABV can be confirmed and thus each isolate should be considered individually concerning its pathogenicity. In contrast to RABV, we could not determine virus shedding in all other lyssavirus-infected mice. This indicates a limited potential of those lyssaviruses to spread beyond the initial spilloverhost, and may explain the absence of onward transmission in non-RABV bat lyssaviruses.

Supporting information

S1 Table. Details of the primer and probe mix for GBLV. The reaction conditions were identical to those used for EBLV-1, BBLV and EBLV-2. (DOCX)

S2 Table. Clinical score sheet for mice, ranging from zero up to five. (DOCX)

S1 Fig. Kaplan-Maier survival plots of the individual isolates following i.c. infection (six Balb/ c mice were inoculated per group) (A) and i.n. infection (three Balb/c mice were inoculated group) (B). Mock-infected control mice did not develop any clinical signs and, hence, were omitted for better visualization. (TIF)

Acknowledgments

The technical assistance by Jeannette Kliemt and Patrick Zitzow is gratefully acknowledged. Also, we would like to thank all animal keepers at FLI for supporting the experimental studies.

Author Contributions

Conceptualization: Thomas Müller.

Formal analysis: Antonia Klein, Madlin Potratz, Luca M. Zaeck, Stefan Finke, Conrad M. Freuling.

Funding acquisition: Thomas Müller, Conrad M. Freuling.

Investigation: Antonia Klein, Elisa Eggerbauer, Madlin Potratz, Luca M. Zaeck, Sten Calvelage, Stefan Finke, Thomas Müller, Conrad M. Freuling.

Methodology: Stefan Finke, Thomas Müller, Conrad M. Freuling.

Resources: Stefan Finke.

Supervision: Thomas Müller.

Visualization: Antonia Klein, Luca M. Zaeck, Conrad M. Freuling.

Writing - original draft: Antonia Klein, Conrad M. Freuling.

Writing – review & editing: Antonia Klein, Elisa Eggerbauer, Madlin Potratz, Luca M. Zaeck, Sten Calvelage, Stefan Finke, Thomas Müller, Conrad M. Freuling.

References

- Amarasinghe GK, Arechiga Ceballos NG, Banyard AC, Basler CF, Bavari S, Bennett AJ, et al. Taxonomy of the order Mononegavirales. update 2018. Arch Virol. 2018 (163): 2283–2294. <u>https://doi.org/10. 1007/s00705-018-3814-x PMID: 29637429</u>
- Fooks AR, Cliquet F, Finke S, Freuling C, Hemachudha T, Mani RS, et al. Rabies. Nat Rev Dis Primers. 2017; 3: 17091. https://doi.org/10.1038/nrdp.2017.91 PMID: 29188797
- Badrane H, Bahloul C, Perrin P, Tordo N. Evidence of Two Lyssavirus Phylogroups with Distinct Pathogenicity and Immunogenicity. J Virol. 2001; 75 (7): 3268–3276. <u>https://doi.org/10.1128/JVI.75.7.3268-</u> 3276.2001 PMID: 11238853
- Markotter W, Coertse J. Bat lyssaviruses. Rev Sci Tech. 2018; 37 (2): 385–400. https://doi.org/10. 20506/rst.37.2.2809 PMID: 30747140
- Nokireki T, Tammiranta N, Kokkonen U-M, Kantala T, Gadd T. Tentative novel lyssavirus in a bat in Finland. Transbound Emerg Dis. 2018; 65: 593–596. https://doi.org/10.1111/tbed.12833 PMID: 29446230
- Coertse J, Grobler C, Sabeta C, Seamark EC, Kearney T, Paweska J, et al. Lyssaviruses in Insectivorous Bats, South Africa, 2003–2018. Emerg Infect Dis. 2020; 26 (12): 3056–3060. <u>https://doi.org/10.3201/eid2612.203592</u> PMID: 33219800
- Müller T, Freuling CM. Rabies in terrestrial animals. In: Fooks AR, Jackson AC, editors. Rabies—Scientific Basis of the Disease and its Management: Elsevier Academic Press. 2020. pp. 195–230.
- Marston DA, Banyard AC, McElhinney LM, Freuling CM, Finke S, Lamballerie X de, et al. The lyssavirus host-specificity conundrum—rabies virus—the exception not the rule. Curr Opin Virol. 2018; 28: 68–73. https://doi.org/10.1016/j.coviro.2017.11.007 PMID: 29182939
- Johnson N, Arechiga-Ceballos N, Aguilar-Setien A. Vampire bat rabies. ecology, epidemiology and control. Viruses. 2014; 6 (5): 1911–1928. https://doi.org/10.3390/v6051911 PMID: 24784570
- Johnson N, Vos A, Freuling C, Tordo N, Fooks AR, Muller T. Human rabies due to lyssavirus infection of bat origin. Vet Microbiol. 2010; 142 (3–4): 151–159. https://doi.org/10.1016/j.vetmic.2010.02.001 PMID: 20188498
- Shinwari MW, Annand EJ, Driver L, Warrilow D, Harrower B, Allcock RJN, et al. Australian bat lyssavirus infection in two horses. Vet Microbiol. 2014; 173 (3–4): 224–231. <u>https://doi.org/10.1016/j.vetmic.</u> 2014.07.029 PMID: 25195190
- Hanlon CA, DeMattos CA, DeMattos CC, Niezgoda M, Hooper DC, Koprowski H, et al. Experimental utility of rabies virus-neutralizing human monoclonal antibodies in post-exposure prophylaxis. Vaccine. 2001; 19 (28–29): 3834–3842. https://doi.org/10.1016/s0264-410x(01)00135-9 PMID: 11427255
- Hanlon CA, Kuzmin IV, Blanton JD, Weldon WC, Manangan JS, Rupprecht CE. Efficacy of rabies biologics against new lyssaviruses from Eurasia. Virus Res. 2005; 111 (1): 44–54. <u>https://doi.org/10.1016/j.virusres.2005.03.009</u> PMID: 15896401
- Malerczyk C, Freuling C, Gniel D, Giesen A, Selhorst T, Müller T. Cross-neutralization of antibodies induced by vaccination with Purified Chick Embryo Cell Vaccine (PCECV) against different Lyssavirus species. Hum Vaccin Immunother. 2014; 10 (10): 2799–2804. https://doi.org/10.4161/21645515.2014. 972741 PMID: 25483634
- 15. Banyard AC, Hayman D, Johnson N, McElhinney L, Fooks AR. Bats and Lyssaviruses. In: Jackson AC, editor. Adv Virus Res. Amsterdam: Elsevier. 2011. pp. 239–289.
- Vos A, Müller T, Cox J, Neubert L, Fooks AR. Susceptibility of ferrets (Mustela putorius furo) to experimentally induced rabies with European Bat Lyssaviruses (EBLV). J Vet Med B Infect Dis Vet Public Health. 2004; 51 (2): 55–60. https://doi.org/10.1111/j.1439-0450.2004.00730.x PMID: 15030601
- Vos A, Müller T, Neubert L, Zurbriggen A, Botteron C, Pöhle D, et al. Rabies in Red Foxes (Vulpes vulpes) Experimentally Infected with European Bat Lyssavirus Type 1. J Vet Med B Infect Dis Vet Public Health. 2004; 51 (7): 327–332. https://doi.org/10.1111/j.1439-0450.2004.00793.x PMID: 15525359
- Constantine DG, Solomon GC. Transmission Experiments with Bat Rabies Isolates. Responses of Certain Carnivores and Rodents to Rabies Viruses from Four Species of Bats. Am J Vet Res. 1968; 29 (1): 181–190. PMID: 5636035
- Cliquet F, Picard-Meyer E, Barrat J, Brookes SM, Healy DM, Wasniewski M, et al. Experimental infection of foxes with European Bat Lyssaviruses type-1 and 2. BMC Vet Res. 2009; 5: 19. <u>https://doi.org/10.1186/1746-6148-5-19 PMID: 19454020</u>
- Vos A, Nolden T, Habla C, Finke S, Freuling CM, Teifke JP, et al. Raccoons (Procyon lotor) in Germany as potential reservoir species for Lyssaviruses. Eur J Wildlife Res. 2013; 59: 637–643. <u>https://doi.org/ 10.1007/s10344-013-0714-y</u>

- Constantine DG. Transmission experiments with bat rabies isolates. reaction of certain Carnivora, opossum, and bats to intramuscular inoculations of rabies virus isolated from free-tailed bats. Am J Vet Res. 1966; 27 (116): 16–19. PMID: 5913022
- Constantine DG. Transmission experiments with bat rabies isolates. responses of certain Carnivora to rabies virus isolated from animals infected by nonbite route. Am J Vet Res. 1966; 27 (116): 13–15. PMID: 5913020
- Perl DF, Bell JF, Moore GJ, Stewart SJ. Chronic recrudescent rabies in a cat. Proc Soc Exp Biol Med. 1977; 155: 540–548. https://doi.org/10.3181/00379727-155-39847 PMID: 896802
- Chen T, Miao FM, Liu Y, Zhang SF, Zhang F, Li N, et al. Possible Transmission of Irkut Virus from Dogs to Humans. Biomed Environ Sci. 2018; 31 (2): 146–148. <u>https://doi.org/10.3967/bes2018.017</u> PMID: 29606193
- Fekadu M, Shaddock JH, Sanderlin DW, Smith JS. Efficacy of rabies vaccines against Duvenhage virus isolated from European house bats (Eptesicus serotinus), classic rabies and rabies-related viruses. Vaccine. 1988; 6 (6): 533–539. https://doi.org/10.1016/0264-410x(88)90107-7 PMID: 3245296
- McColl KA, Chamberlain T, Lunt RA, Newberry KM, Middleton D, Westbury HA. Pathogenesis studies with Australian bat lyssavirus in grey-headed flying foxes (Pteropus poliocephalus). Aust Vet J. 2002; 80 (10): 636–641. https://doi.org/10.1111/j.1751-0813.2002.tb10973.x PMID: 12465817
- Murphy FA, Bell JF, Bauer SP, Gardner JJ, Moore GJ, Harrison AK, et al. Experimental chronic rabies in the cat. Lab Invest. 1980; 43 (3): 231–241. PMID: 6995713
- Trimarchi CV, Rudd RJ, Abelseth MK. Experimentally induced rabies in four cats inoculated with rabies virus isolated from a bat. Am J Vet Res. 1986; 47 (4): 777–780. PMID: 3516030
- 29. Fekadu M. Pathogenesis of Rabies Virus Infection in Dogs. Rev Infect Dis. 1988; 10 (4): 678–683. https://doi.org/10.1093/clinids/10.supplement_4.s678 PMID: 3206080
- McColl KA, Chamberlain T, Lunt RA, Newberry KM, Westbury HA. Susceptibility of domestic dogs and cats to Australian bat lyssavirus (ABLV). Vet Microbiol. 2007; 123 (1–3): 15–25. <u>https://doi.org/10.1016/j.vetmic.2007.03.024</u> PMID: 17449202
- Tignor GH, Shope RE, Bhatt PN, Percy DH. Experimental Infection of Dogs and Monkeys with 2 Rabies Serogroup Viruses, Lagos Bat and Mokola (Iban 27377)—Clinical, Serologic, Virologic, and Fluorescent-Antibody Studies. J Infect Dis. 1973; 128 (4): 471–478. <u>https://doi.org/10.1093/infdis/128.4.471</u> PMID: 4200590
- Percy DH, Bhatt PN, Tignor GH, Shope RE. Experimental Infection of Dogs and Monkeys with 2 Rabies Serogroup Viruses, Lagos Bat and Mokola (Iban 27377)—Gross Pathologic and Histopathologic Changes. Vet Pathol. 1973; 10 (6): 534–549. https://doi.org/10.1177/030098587301000607 PMID: 4214315
- Charlton KM, Casey GA, Webster WA, Bundza A. Experimental rabies in skunks and foxes. Pathogenesis of the spongiform lesions. Lab Invest. 1987; 57 (6): 634–645. PMID: 3695410
- Eggerbauer E, Pfaff F, Finke S, Hoper D, Beer M, Mettenleiter TC, et al. Comparative analysis of European bat lyssavirus 1 pathogenicity in the mouse model. PLoS Neglect Trop Dis. 2017; 11 (6): e0005668. https://doi.org/10.1371/journal.pntd.0005668 PMID: 28628617
- Markotter W, Kuzmin IV, Rupprecht CE, Nel LH. Lagos bat virus virulence in mice inoculated by the peripheral route. Epidemiol Infect. 2009; 137 (8): 1155–1162. <u>https://doi.org/10.1017/</u> S0950268808001945 PMID: 19144249
- 36. Healy DM, Brookes SM, Banyard AC, Núñez A, Cosby SL, Fooks AR. Pathobiology of rabies virus and the European bat lyssaviruses in experimentally infected mice. Virus Res. 2013; 172 (1–2): 46–53. https://doi.org/10.1016/j.virusres.2012.12.011 PMID: 23274107
- Hicks DJ, Nunez A, Healy DM, Brookes SM, Johnson N, Fooks AR. Comparative Pathological Study of the Murine Brain after Experimental Infection with Classical Rabies Virus and European Bat Lyssaviruses. J Comp Pathol. 2009; 140 (2–3): 113–126. <u>https://doi.org/10.1016/j.jcpa.2008.09.001</u> PMID: 19111840
- Kgaladi J, Nel LH, Markotter W. Comparison of pathogenic domains of rabies and African rabies-related lyssaviruses and pathogenicity observed in mice. Onderstepoort J Vet Res. 2013; 80 (1): E1–E13. https://doi.org/10.4102/ojvr.v80i1.511 PMID: 23718883
- Banyard AC, Healy DM, Brookes SM, Voller K, Hicks DJ, Nunez A, et al. Lyssavirus infection. 'Iow dose, multiple exposure' in the mouse model. Virus Res. 2014; 181: 35–42. <u>https://doi.org/10.1016/j.virusres.2013.12.029</u> PMID: 24380842
- 40. Nolden T, Banyard AC, Finke S, Fooks AR, Hanke D, Höper D, et al. Comparative studies on genetic, antigenic and pathogenic characteristics of Bokeloh bat lyssavirus. J Gen Virol. 2014; 95: 1647–1653. https://doi.org/10.1099/vir.0.065953-0 PMID: 24828330

- Mollentze N, Biek R, Streicker DG. The role of viral evolution in rabies host shifts and emergence. Curr Opin Virol. 2014; 8: 68–72. https://doi.org/10.1016/j.coviro.2014.07.004 PMID: 25064563
- 42. Potratz M, Zaeck L, Christen M, Te Kamp V, Klein A, Nolden T, et al. Astrocyte Infection during Rabies Encephalitis Depends on the Virus Strain and Infection Route as Demonstrated by Novel Quantitative 3D Analysis of Cell Tropism. Cells. 2020; 9 (2). https://doi.org/10.3390/cells9020412 PMID: 32053954
- Guillen J. FELASA guidelines and recommendations. J Am Assoc Lab Anim Sci. 2012; 51 (3): 311– 321. PMID: 22776188
- Calvelage S, Freuling CM, Fooks AR, Höper D, Marston DA, McElhinney L, et al. Full-Genome Sequences and Phylogenetic Analysis of Archived Danish European Bat Lyssavirus 1 (EBLV-1) Emphasize a Higher Genetic Resolution and Spatial Segregation for Sublineage 1a. Viruses. 2021; 13 (4). https://doi.org/10.3390/v13040634 PMID: 33917139
- **45.** Kärber G. Beitrag zur kollektiven Behandlung pharmakologischer Reihenversuche. Arch Exp Path Pharmakol. 1931; 162: 480–483.
- Dean DJ, Abelseth MK, Athanasiu P. The fluorescence antibody test. In: Meslin FX, Kaplan MM, Koprowski H, editors. Laboratory techniques in rabies. Geneva, Switzerland: World Health Organization. 1996. pp. 88–93.
- Hoffmann B, Freuling CM, Wakeley PR, Rasmussen TB, Leech S, Fooks AR, et al. Improved Safety for Molecular Diagnosis of Classical Rabies Viruses by Use of a TaqMan Real-Time Reverse Transcription-PCR "Double Check" Strategy. J Clin Microbiol. 2010; 48 (11): 3970–3978. <u>https://doi.org/10.1128/ JCM.00612-10 PMID: 20739489</u>
- Fischer M, Freuling CM, Müller T, Wegelt A, Kooi EA, Rasmussen TB, et al. Molecular double-check strategy for the identification and characterization of European Lyssaviruses. J Virol Methods. 2014; 203: 23–32. https://doi.org/10.1016/j.jviromet.2014.03.014 PMID: 24681051
- Webster WA, Casey GA. Virus isolation in neuroblastoma cell culture. In: Meslin FX, Kaplan MM, Koprowski H, editors. Laboratory techniques in rabies. Geneva, Switzerland: World Health Organization. 1996. pp. 93–104.
- 50. Orbanz J, Finke S. Generation of recombinant European bat lyssavirus type 1 and inter-genotypic compatibility of lyssavirus genotype 1 and 5 antigenome promoters. Arch Virol. 2010; 155 (10): 1631–1641. https://doi.org/10.1007/s00705-010-0743-8 PMID: 20614145
- Renier N, Wu Z, Simon DJ, Yang J, Ariel P, Tessier-Lavigne M. iDISCO: a simple, rapid method to immunolabel large tissue samples for volume imaging. Cell. 2014; 159 (4): 896–910. <u>https://doi.org/10.1016/j.cell.2014.10.010 PMID: 25417164</u>
- 52. Grüneboom A, Hawwari I, Weidner D, Culemann S, Müller S, Henneberg S, et al. A network of transcortical capillaries as mainstay for blood circulation in long bones. Nat Metab. 2019; 1 (2): 236–250. https://doi.org/10.1038/s42255-018-0016-5 PMID: 31620676
- 53. Klingberg A, Hasenberg A, Ludwig-Portugall I, Medyukhina A, Männ L, Brenzel A, et al. Fully Automated Evaluation of Total Glomerular Number and Capillary Tuft Size in Nephritic Kidneys Using Lightsheet Microscopy. J Am Soc Nephrol. 2017; 28 (2): 452–459. <u>https://doi.org/10.1681/ASN.2016020232</u> PMID: 27487796
- Zaeck LM, Scheibner D, Sehl J, Müller M, Hoffmann D, Beer M, et al. Light Sheet Microscopy-Assisted 3D Analysis of SARS-CoV-2 Infection in the Respiratory Tract of the Ferret Model. Viruses. 2021; 13 (3). https://doi.org/10.3390/v13030529 PMID: 33807059
- Zaeck L, Potratz M, Freuling CM, Müller T, Finke S. High-Resolution 3D Imaging of Rabies Virus Infection in Solvent-Cleared Brain Tissue. J Vis Exp. 2019 (146). <u>https://doi.org/10.3791/59402</u> PMID: 31107452
- 56. Schindelin J, Arganda-Carreras I, Frise E, Kaynig V, Longair M, Pietzsch T, et al. Fiji: an open-source platform for biological-image analysis. Nat Methods. 2012; 9 (7): 676–682. <u>https://doi.org/10.1038/nmeth.2019 PMID: 22743772</u>
- Bolte S, Cordelières FP. A guided tour into subcellular colocalization analysis in light microscopy. J Microsc. 2006; 224 (3): 213–232. https://doi.org/10.1111/j.1365-2818.2006.01706.x PMID: 17210054
- Office International des Epizooties. Newcastle disease (infection with newcastle disease virus). OIE terrestrial Manual. 2012; Chapter 3.3.14. Available: https://www.oie.int/fileadmin/Home/eng/Health_ standards/tahm/3.03.14_NEWCASTLE_DIS.pdf. Accessed July 20, 2021.
- 59. George DB, Webb CT, Farnsworth ML, O'Shea TJ, Bowen RA, Smith DL, et al. Host and viral ecology determine bat rabies seasonality and maintenance. Proc Natl Acad Sci U S A. 2011; 108 (25): 10208–10213. https://doi.org/10.1073/pnas.1010875108 PMID: 21646516
- Blackwood JC, Streicker DG, Altizer S, Rohani P. Resolving the roles of immunity, pathogenesis, and immigration for rabies persistence in vampire bats. Proc Natl Acad Sci U S A. 2013; 110 (51): 20837– 20842. https://doi.org/10.1073/pnas.1308817110 PMID: 24297874

- Pajamo K, Harkess G, Goddard T, Marston D, McElhinney L, Johnson N, et al. Isolation of European bat lyssavirus type 2 (EBLV-2) in a Daubenton's bat in the UK with a minimum incubation period of 9 months. Rabies Bull Eur. 2008; 32 (2): 6–7.
- Moore GJ, Raymond GH. Prolonged incubation period of rabies in a naturally infected insectivorous bat, Eptesicus fuscus. J Wildl Dis. 1970; 6 (3): 167–168. https://doi.org/10.7589/0090-3558-6.3.167 PMID: 5424435
- Freuling C, Vos A, Johnson N, Fooks A, Müller T. Bat rabies—a Gordian knot? Berl Munch Tierarztl Wochenschr. 2009; 122 (11/12): 425–433.
- Liu Y, Chen Q, Zhang F, Zhang S, Li N, Lian H, et al. Evaluation of rabies biologics against Irkut virus isolated in China. J Clin Microbiol. 2013; 51 (11): 3499–3504. https://doi.org/10.1128/JCM.01565-13 PMID: 23946522
- Regnault B, Evrard B, Plu I, Dacheux L, Troadec E, Cozette P, et al. First case of lethal encephalitis in Western Europe due to European bat lyssavirus type 1. Clin Infect Dis. 2021. <u>https://doi.org/10.1093/ cid/ciab443</u> PMID: 33991184
- Selimov MA, Tatarov AG, Botvinkin AD, Klueva EV, Kulikova LG, Khismatullina NA. Rabies-related Yuli virus; identification with a panel of monoclonal antibodies. Acta Virol. 1989; 33 (6): 542–546. PMID: 2576595
- Velasco-Villa A, Mauldin MR, Shi M, Escobar LE, Gallardo-Romero NF, Damon I, et al. The history of rabies in the Western Hemisphere. Antiviral Res. 2017; 146: 221–232. <u>https://doi.org/10.1016/j.</u> antiviral.2017.03.013 PMID: 28365457
- Banyard AC, Davis A, Gilbert AT, Markotter W. Bat rabies. In: Fooks AR, Jackson AC, editors. Rabies —Scientific Basis of the Disease and its Management: Elsevier Academic Press. 2020. pp. 231–276.
- 69. Fooks AR, Jackson AC. 2020. Rabies—Scientific Basis of the Disease and its Management: Elsevier Academic Press.
- Gibbons RV. Cryptogenic rabies, bats, and the question of aerosol transmission. Ann Emerg Med. 2002; 39 (5): 528–536. https://doi.org/10.1067/mem.2002.121521 PMID: 11973559
- **71.** Lafay F, Coulon P, Astic L, Saucier D, Riche D, Holley A, et al. Spread of the CVS strain of rabies virus and of the avirulent mutant AvO1 along the olfactory pathways of the mouse after intranasal inoculation. Virology. 1991; 183 (1): 320–330. https://doi.org/10.1016/0042-6822(91)90145-2 PMID: 2053286
- 72. Freuling C, Vos A, Johnson N, Kaipf I, Denzinger A, Neubert L, et al. Experimental infection of serotine bats (Eptesicus serotinus) with European bat lyssavirus type 1a. J Gen Virol. 2009; 90 (10): 2493–2502. https://doi.org/10.1099/vir.0.011510-0 PMID: 19515825
- Johnson N, Vos A, Neubert L, Freuling C, Mansfield KL, Kaipf I, et al. Experimental study of European bat lyssavirus type-2 infection in Daubenton's bats (Myotis daubentonii). J Gen Virol. 2008; 89 (11): 2662–2672. https://doi.org/10.1099/vir.0.2008/003889-0 PMID: 18931061
- 74. Russell WMS, Burch LR. The principles of humane experimental technique. London: Universities Federation for Animal Welfare. 1959.
- 75. Troupin C, Dacheux L, Tanguy M, Sabeta C, Blanc H, Bouchier C, et al. Large-Scale Phylogenomic Analysis Reveals the Complex Evolutionary History of Rabies Virus in Multiple Carnivore Hosts. PLoS Pathog. 2016; 12 (12): e1006041. https://doi.org/10.1371/journal.ppat.1006041 PMID: 27977811
- 76. Troupin C, Picard-Meyer E, Dellicour S, Casademont I, Kergoat L, Lepelletier A, et al. Host Genetic Variation Does Not Determine Spatio-Temporal Patterns of European Bat 1 Lyssavirus. Genome Biol Evol. 2017; 9 (11): 3202–3213. https://doi.org/10.1093/gbe/evx236 PMID: 29165566
- Sanjuán R, Domingo-Calap P. Mechanisms of viral mutation. Cell Mol Life Sci. 2016; 73 (23): 4433– 4448. https://doi.org/10.1007/s00018-016-2299-6 PMID: 27392606